Author’s response to reviews

Title: Viscoelastic properties of bovine knee joint articular cartilage: dependency on thickness and loading frequency.

Authors:

Daniel M Espino (d.m.espino@bham.ac.uk)
Duncan E.T. Shepherd (d.e.shepherd@bham.ac.uk)
David W.L. Hukins (d.w.hukins@bham.ac.uk)

Version: 2 Date: 23 July 2013

Author’s response to reviews: see over
RE: Viscoelastic properties of bovine knee joint articular cartilage: dependency on thickness and loading frequency.

Dear Editor,

Please find enclosed an original article being submitted for publication to BMC Musculoskeletal Disorders. This manuscript describes changes in the dynamic mechanical behaviour of articular cartilage under physiological and traumatic loading rates. This is of importance to knee joint function and has possible consequences for osteoarthritis.

In submitting the above manuscript I state that there has been no duplicate publication or submission of the work described in this manuscript elsewhere. All authors have read and approved the manuscript, if accepted the authors will transfer copyright to the publisher. No conflict of interests exist for the publication of this manuscript. All sources of funding have been acknowledged.

Yours Sincerely,

Daniel M Espino
BSc, PhD
Lecturer in Bio-medical Engineering